

MDPI

Article

Does Firm Political Risk Affect the Relationship between Corporate Social Responsibility and Firm Value?

Ahmed Marhfor 1,2,*, Kais Bouslah 2,3 and Abdelmajid Hmaittane 2,4

- ¹ Management Science Department, Université du Québec en Abitibi-Témiscamingue, Rouyn-Noranda, QC J9X 5E4, Canada
- ² African Chair of Innovation and Sustainable Management (CAIMD-UM6P), Mohamed VI Polytechnic University, Ben Guerir 43150, Morocco
- ³ Centre for Responsible Banking and Finance, School of Management, University of St Andrews, North Haugh KY16 9RJ, UK
- ⁴ National School of Trade & Management (ENCG), Ibn Zohr University, Agadir 80000, Morocco
- * Correspondence: ahmed.marhfor@uqat.ca

Abstract: This paper investigates whether firm-level (idiosyncratic) political risk (PR) affects the relationship between corporate social responsibility (CSR) and firm value using a sample of 16,518 firm-year observations which correspond to 2055 unique firms belonging to the Russell 3000 Index over the sample period 2010–2020. Our main findings are as follows: First, firm-level PR does not affect firm value. Second, CSR is positively related to firm value, which is mainly driven by the social component of CSR. Finally, PR has no effect on the CSR-firm value relationship, regardless of the PR type. Our evidence suggests that firm-level PR is not priced in the financial market and as such it does not affect the CSR-firm value relationship. This is consistent with portfolio theory which suggests that only systematic risk is priced.

Keywords: firm-level political risk; corporate social responsibility; firm value; systematic political risk

Citation: Marhfor, A.; Bouslah, K.; Hmaittane, A. Does Firm Political Risk Affect the Relationship between Corporate Social Responsibility and Firm Value? *Sustainability* **2022**, *14*, 11217. https://doi.org/10.3390/ su141811217

Academic Editor: Gaetano della Corte

Received: 28 July 2022 Accepted: 31 August 2022 Published: 7 September 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

1. Introduction

During the last two decades, the concept of corporate social responsibility (CSR) is gradually gaining importance among corporations, investors, and policymakers. This increased attention is due, among other things, to factors related to corporate behaviors (e.g., corporate scandals, 2000 dot-com bubble, 2008 financial crisis) and investor preferences regarding social, environmental, and governance (ESG) issues. The signing of the US climate law, the Inflation Reduction Act, by U.S. President Joe Biden on 16 August 2022 is a good example of this increased attention from regulators [1]. The new climate law allocates \$370 billion (e.g., tax credits and tax credit bonuses) to decarbonization efforts and is expected to enable the U.S. to cut its emissions by 40% by 2030 relative to 2005 levels [1]. The signing in 2006 of the UN Principle for Responsible Investment (PRI) and its evolution is another good example of this increased attention from investors. CSR has therefore become an important issue for firms and the subject of an intense debate among academics, practitioners, and policymakers.

The CSR literature provides mixed views regarding the theoretical underpinning and financial implications of CSR. Three theoretical arguments most cited are shareholder theory, agency theory, and stakeholder theory. The shareholder theory suggests that managers should maximize shareholders' wealth and consider CSR to be an additional cost that reduces profits [2]. Agency theory suggests that managers could implement CSR actions opportunistically to extract personal benefits at the expense of other stakeholders, including shareholders [3–5]. The stakeholder theory suggests that managers should maximize stakeholders' welfare and consider CSR to be an investment that increases profits [6–9].

Sustainability **2022**, 14, 11217 2 of 24

Much of the CSR literature focuses on the relationship between CSR and firm value or financial performance [10,11]. However, the evidence is mixed. Several studies find a positive relation [12–17], whereas other studies do not find any relation [18–20] between CSR and firm value or financial performance. Other studies find a nonlinear relationship or provide an alternative explanation to CSR, e.g., CSR signals private information about future performance [21,22]. Despite the mixed evidence, several studies show various channels through which CSR can enhance firm value, including increased future revenue growth [11], lower cost of capital, higher institutional ownership and analyst coverage [23], increased transparency [24], better access to finance [25], and lower risk [26].

The mixed evidence regarding the financial implications of CSR led to an important research stream in the CSR literature focusing on the roles of variables moderating or conditioning this relationship [27–30]. That is, CSR can affect firm value only under certain conditions. For example, ref. [27] showed that CSR and firm value are positively related only for firms with higher customer awareness. Ref. [28] found that the impact of CSR varies with the level of influential institutional ownership and depends upon economic conditions (e.g., 2008–2009 financial crisis). Ref. [31] found that large shareholder ownership lowers the relation between CSR and firm value in Korea. The authors of [32] reported similar findings in Turkey. In [33], the authors found that CSR activities had a positive effect on returns only during the 2008-2009 financial crisis. More recently, some studies have shown that political risk (PR) could affect the CSR–firm value relationship [34,35]. Ref. [35] found that politically connected firms overinvest in CSR to obtain political legitimacy and access to governmental resources. Our paper contributes to this research stream by examining whether *firm-specific* PR moderates the CSR–firm value relationship.

Politicians and regulatory bodies frequently shift their policies, causing a high degree of PR. The overall uncertainty attributed to the political system can hamper value creation because it creates difficulties for firms regarding funding, planning, and maximizing the outcomes of their activities. For instance, ref. [36] found that PR depresses firm investment. PR can increase the cost of capital [37], return volatility [38], analyst forecast error [39], bid-ask spread [40], stock price crash risk [41], earnings management [42], and corporate risk-taking [43]; it can also reduce stock prices [44], merger-and-acquisition (M&A) deals [45], and IPO activity [46]. Moreover, ref. [47] found a negative relation between politically connected firms and analysts' stock recommendations. Nevertheless, some firms and industries have proven to be more resilient than others in preserving their value during periods of high PR. For instance, ref. [48] showed that high PR has a negative valuation effect in only 33% of Fama-French 30 industries while investment has decreased in 90% of them. On the other hand, 3% of Fama-French 30 industries had higher valuation and 64% of the industries were not affected. This narrow impact on market value is quite surprising [48]. Previous literature focused on the effects of aggregate levels of PR (instead of firm-level PR) on a variety of corporate decisions and asset pricing e.g., [44,49]. However, our understanding of the potential effects of firm-level PR is still limited.

This paper examines whether firm-specific PR moderates the CSR-firm value relationship. Our paper is related to the work of [33] and [34]. Ref. [33] showed that CSR activities had a positive effect on returns only during the 2008–2009 financial crisis. Ref. [34] found that there is a positive association between CSR and firm value only when aggregate (systematic) PR is high. These findings suggest that CSR engagement is not relevant during "normal" and more "steady-state" times. In other words, CSR activities can add value to firms only under certain circumstances (e.g., severe PR).

Based on the stakeholder theory, we argue that CSR activities can preserve firm value when firm-specific PR is high. Strong CSR engagement should allow firms to produce and maintain high levels of social capital (e.g., trust and reputation) that can be used to sustain their competitive advantage during times of severe PR [33,34]. In the same line of reasoning, PR can negatively affect the quality of public information, leading to high demand for additional firm-level information [8,32]. The disclosure of CSR-related information can reduce information asymmetry between the firm and its stakeholders (including

Sustainability **2022**, 14, 11217 3 of 24

shareholders) and ultimately offset the negative consequences of information ambiguity and opacity during periods of severe PR [39,40]. Managers can avoid altering their risk-management strategies during periods of high PR because CSR can reduce the cost of capital [50,51] and risk [52].

Our paper is closely related to the work of [34] which examines the impact of aggregate (systematic) PR on the CSR–firm value relationship. Our analysis differs from theirs in two ways. First, ref. [34] uses the index from [49] as a proxy for aggregate (systematic) PR, while our tests rely on firm-specific measures of PR developed by [53]. Ref. [53] applies text-searching techniques to construct a new measure of firm-specific PR which differs from existing measures in several aspects. For instance, the proposed PR measure indicates the share of firm quarterly earnings conferences devoted to PR. It reflects the perception of managers and participants in the conference calls. Moreover, most of the variation in the measure is at the firm level. For instance, "only a very small number of firms involved with power generation will be affected by new regulations governing the emissions of mercury from coal furnaces across state lines or changing rules about the compensation for providing spare generation capacity in Ohio" [53] (pp. 5). Second, we also examine the eight different channels (components) of firm-specific PR, stemming from economic, environment, health, institutions, tax, technology, trade, and security issues, as some channels may have stronger, weaker, or neutral effects on the CSR–firm value relationship.

Our findings show that firms with strong CSR engagement perform better during times of high PR as well as during "normal" and more "steady-state" times. However, we find that PR has a neutral impact on the CSR-firm value relationship. This neutral effect holds when we examine the eight category-specific indexes of PR provided by [53], when we partition our sample based on nine industries (communications, consumer discretionary, consumer staples, energy, health care, industrials, materials, real estate, and technology), and when we decompose CSR activities into three subcategories (Environment, Social, and Governance). It appears that some firms in each industry cope better with high PR while others are unable to preserve their value, so that on average the effect of PR is neutral. Furthermore, being idiosyncratic in nature, firm-level PR may be fully diversifiable, and only PR at the aggregate level is priced. This may explain why we find a neutral impact of PR while prior literature shows significant valuation effects based on systematic PR.

The remainder of this paper is structured as follows: Section 2 summarizes the literature on the impact of PR and CSR on firm value, and discusses the theoretical arguments behind our hypotheses. Section 3 describes our research design and variable construction. Section 4 presents the findings and Section 5 discusses these findings and their implications. Section 6 concludes the paper.

2. Review of Literature and Hypotheses Development

Our research is related to two important strands of the literature. The first literature strand examines the effects of PR on firm's decisions and outcomes. The second literature strand examines the impact of CSR on firm value. The first and second parts of this section discuss these two streams, respectively. The last part of this section discusses the potential moderating role of PR in the CSR–firm value relationship.

2.1. The valuation effects of PR

Numerous studies in the literature examine the relationship between PR and corporate investment [36]; M&A activities [45]; cost of external financing [54,55]; stock price volatility [44]; debt maturity and leverage [56]; IPO activities [46]; accounting conservatism [57]; earnings management [42]; analyst forecast errors [39]; and analysts' stock recommendations [47]. In general, two channels through which PR can affect firm value are proposed. The first channel focuses largely on the real option theory which predicts that high levels of PR will increase the value of the option to delay a firm's investments until the uncertainty starts to subside [36,45]. The option to wait should be stronger for more

Sustainability **2022**, 14, 11217 4 of 24

irreversible investments and weaker for investments that are most costly (or difficult) to delay [36,45]. Consistent with these predictions, ref. [36] found that the decrease in corporate investment during times of high PR is stronger for firms with more irreversible investments. The research found in [45] has shown that PR has a more pronounced impact on M&A deals in industries with high concentration or low merger activity, where competition is weak and M&A activity can be postponed more easily.

Moving away from the real option channel, a growing number of studies rely on the information channel to explain the detrimental effects of PR. As suggested by [55], a firm's production, hiring, financing, and investment activities transmit valuable information to different market participants. During high-PR periods, a slowdown in these activities may decrease the amount of information that market participants receive from the firm. This could lead to severe agency problems between stakeholders and managers, resulting in distortions in allocative efficiency [48]. High PR could also have a detrimental effect on financial analysts' role in transmitting firm-level information to market participants. In this respect, ref. [39] has shown that the accuracy of analysts' forecasts is lower when PR is high. In the same line of reasoning, when PR increases market participants' uncertainty with respect to firms' prospects, this could have a negative impact on managers' learning capacities.

The managerial learning hypothesis predicts that managers can learn new information from a firm's stock prices and use this information when making investment decisions [58–60]. We argue that informed investors' trades can impound information into stock prices that is new to managers, i.e., managers do not have such information. For example, managers can learn new information from policymakers, institutional investors, customers, and financial analysts about the future state of the economy, the demand for a firm's products, and the competition with other firms [61]. The learning process should allow managers to add new information to their own information and make value-maximizing investment decisions [61]. If PR impacts managerial learning negatively, high levels of PR may distort the fundamental role of markets in guiding managers to make efficient investment decisions (high revelatory efficiency) [55]. Consistent with these predictions, ref. [55] shows that the sensitivity of investment to a given firm's cost of capital decreases when PR is high. In the same line of reasoning, ref. [48] also found that investment–price sensitivity is low during times of high PR. The above arguments lead to the following hypothesis:

Hypothesis 1: *There is a negative relationship between PR and firm value.*

A less-explored channel in the PR literature is the managerial risk-taking incentives. We argue that managers' risk-taking behavior can play an important role in investment decisions. In this respect, adequate risk-management strategies during high-PR periods can help firms avoid "negative" uncertainty while making use of "positive" uncertainty [43,62,63]. We argue that managers who are unable to differentiate between "negative" and "positive" uncertainty will reduce investment expenditures and take a defensive position. Such typical risk-avoiding behavior should not allow managers to support high-risk and high-yield investment opportunities [43,62,63]. In this respect, ref. [62] has shown that when firms cannot fully insure against PR, they have incentives to reduce capital stock. On the other hand, if firms can identify the potential opportunities of future PR uncertainty, they should be able to support higher risk associated with higher-yield investment projects and ultimately preserve value creation during times of high PR [43,63]. This leads to the following hypothesis:

Hypothesis 2: Well-designed risk-taking incentives under high PR can preserve firm value.

Sustainability **2022**, 14, 11217 5 of 24

The CSR literature provides mixed views regarding the theoretical underpinning, definition, measurement, integration and implementation, and financial implications of CSR. CSR literature provides at least three theoretical arguments underpinning CSR: shareholder theory, agency theory and stakeholder theory. The shareholder theory suggests that managers should maximize shareholders' wealth and consider CSR to be an additional cost that reduces profits and consequently firm value [2]. Agency theory suggests that managers could act opportunistically to extract personal benefits at the expense of other stakeholders, including shareholders. In this case, managers implement CSR actions to build their personal reputation as good/proactive citizens, i.e., career concerns, e.g., to easily find a job elsewhere if fired or justify poor financial performance by the firm's engagement in CSR actions [3-5]. The stakeholder theory suggests that managers should maximize stakeholders' welfare and consider CSR as an investment that increases profits and consequently firm value [6,7,9]. Furthermore, the stakeholder theory provides two CSR versions: normative CSR and instrumental or strategic CSR [8]. Normative CSR is grounded in ethical and moral values (i.e., CSR actions are the right things to do based on ethical or moral values irrespective of their financial implications). Instrumental or strategic CSR suggests that a firm implements CSR for strategic reasons, e.g., as a response to stakeholders' demands or expectations. For instance, firms could be reactive to stakeholders' demands in order to obtain resources or their support, either to continue to do business (social license to operate) or to avoid ESG risks (e.g., avoid negative media or publicity that could ultimately affect the survival of the firm).

The CSR literature also provides mixed views regarding CSR meaning (definition, form, or type) and measurement, i.e., what to include or exclude [64,65]. Ref. [66] argues that the CSR concept is largely debated among academics and practitioners, and applied differently both across countries and across firms within the same country or industry. CSR operationalization (i.e., measurement) also differs between and within academics and practitioners. The proliferation of ESG data providers and the variety of these data illustrate the difficulties associated with CSR operationalization. Another dilemma is whether one should focus on an aggregated CSR measure or the individual CSR components, which themselves have subcomponents that vary in intensity and importance over time for the same firm and across firms and industries.

An important emerging research stream in the CSR literature focuses on CSR integration and implementation into corporate identity and strategies. Ref. [65] proposes a model of CSR development and implementation including seven organizational stages articulated around three cultural phases (CSR reluctance, CSR grasp, and CSR embedment). The CSR reluctance phase includes one dismissing stage (CSR is perceived as a constraint); the CSR grasp phase includes a self-protecting stage, a compliance-seeking stage, and a capability-seeking stage; and the CSR embedment phase includes a caring stage, a strategizing stage, and a transforming stage where CSR becomes fully integrated into every aspect of the firm's activities. The support of the organizational culture or corporate identity (values, beliefs, attitudes and practices) is essential for the model in [65] to be successful. That is, CSR integration involves the integration of CSR principles into corporate identity so that the firm makes the transition to positive corporate identity where the focus is on positive outcomes, processes, and attributes, e.g., moving from harm avoidance and focusing on self-interest to openness and helping others [66,67]. Ref. [66] argues that corporate values are the cornerstone of both positive corporate identity and CSR. Applying the model of [65] to a case study, they found higher inconsistencies between real corporate values (values driving the firm's operating activity), ideal corporate values (values driving the organization), and CSR values (firm's CSR principles), which has led to a lower CSR integration into corporate identity. To achieve its full potential and be communicated efficiently to outsiders, CSR needs to be fully integrated into corporate identity [66]. This is consistent with [68] who argue that building a focused and integrated CSR as part of a firm's core strategies can lead to a sustainable competitive advantage. Ref. [69] examined the CSR structures of U.S. firms and found evidence of CSR specialization, Sustainability **2022**, 14, 11217 6 of 24

where most firms focus on a single CSR dimension. They also find that CSR diversification had a positive impact on firm value before and during the 2008 financial crisis.

Much of the CSR literature strives to establish a business case for CSR [10,11]. As such, the large body of CSR literature focuses on the financial implications of CSR. However, this literature is characterized by the absence of consensus regarding the relationship between CSR and firm value or financial performance. The evidence is mixed. Numerous studies have reported a positive relation between CSR and firm value or financial performance [12–17]. Other studies have not found any relation between CSR and firm value or financial performance [18–20]. In addition, ref. [22] found an inverted U-shaped relationship between CSR and firm value in mainland China and Hong Kong. Ref. [21] provides an alternative explanation to the positive relation between CSR and firm value. They showed that CSR signals private information about future performance. In other words, high-CSR firms are those expecting higher future performance.

In sum, the CSR-firm value relationship has been extensively examined in the literature with no conclusive results. Nonetheless, several studies report evidence of various channels through which CSR can enhance firm value, including increased future revenue growth [11], lower cost of capital, higher institutional ownership and analyst coverage [23], increased transparency [24], better access to finance [25], and lower risk [26]. Ref. [11] found that philanthropy increases future revenue growth through attracting and retaining customers (i.e., increased name and brand recognition). Ref. [25] found that CSR leads to better access to finance because of lower agency costs (better engagement with stakeholders) and increased transparency (lower informational asymmetry).

Under the stakeholder theory, CSR engagement is viewed as a way to resolve conflicts among stakeholders and run the firm to the mutual advantage of its stakeholders, including the firm's shareholders [34,70,71]. A higher protection of stakeholders' interests could help firms build forms of social capital such as legitimacy, reputation, and trust [34]. As a result, stakeholders (e.g., employees, customers, suppliers, and communities) should perceive the firm as an attractive partner [71]. Firms should, then, profit from more motivated employees and loyal customers [34,71]. Suppliers may also engage in lowering firms' costs, increasing quality, and investing in strategic and innovative relationships with the firm. In the same line of reasoning, financial partners can also reciprocate CSR engagement and lower the cost of capital [50,51]. Under the stakeholder theory, CSR activities can also serve as a powerful mechanism that reduces firm risk and agency costs [34,70,71]. The above arguments lead to the following hypothesis:

Hypothesis 3: *There is a positive relationship between CSR and firm value.*

2.3. The moderating role of firm-level PR

The mixed evidence regarding the signs and direction of a relationship between CSR and firm value or financial performance has led to an important research stream in the CSR literature focusing on the roles of variables moderating or conditioning this relationship. A growing number of studies suggest that the focus on the direct relationship between CSR and firm value can lead to spurious and imprecise findings because many influential factors can indirectly affect the potential CSR-firm value relationship [27–30]. For instance, ref. [27] has shown that CSR and firm value are positively related only for firms with higher customer awareness. In [28], the research has shown that the CSR-firm value relationship is affected by influential institutional ownership and economic conditions (e.g., 2008 financial crisis). The authors of [31] found that large shareholder ownership lowers the relation between CSR and firm value in Korea. They concluded that ownership structure moderates the CSR-firm value relationship. Ref. [32] reports similar findings in Turkey. Ref. [33] found that CSR activities have a positive effect on returns only during the 2008-2009 financial crisis. The findings of ref. [29] suggest that the CSR-firm value relationship is moderated by firm size and age. In [34], the authors also found that the CSR-firm value relationship is moderated by PR. This new line of research suggests Sustainability **2022**, 14, 11217 7 of 24

that CSR is value-relevant only under certain conditions. Our paper contributes to this research stream by examining whether firm-specific PR moderates the CSR-firm value relationship.

We argue that CSR activities can decrease the value of the option to delay a firm's investment. During times of high PR, loyal customers may maintain their demand and even accept paying higher prices. As a result, uncertainty related to project revenues should be less severe for firms with strong CSR engagement. At the same time, suppliers may also support the firm and act as a buffer against potential changes in a firm's costs by maintaining such costs at low levels and engaging in innovative relationships with the firm. This will reduce the uncertainty related to project costs and ultimately reduce the option to delay a firm's investments. In addition, ref. [34] argues that firms with strong CSR engagement should be able to obtain financial resources at more favorable terms when PR is high because of greater stakeholder support, enhanced firm reputation, and trust. Lower financing constraints should also decrease the incentive to delay a firm's projects and offset the negative consequences linked to PR.

CSR engagement can reduce information asymmetry between managers and firms' stakeholders [39,72], and signal a higher quality of firms' future prospects [73–75]. We expect these effects to be stronger when PR is high. The research in [39] documents the fact that good CSR practices moderate analyst forecast error when PR is high. Ref. [61] argues that managers can learn from informed stakeholders and adjust their CSR practices accordingly. This could enhance their ability to make value-maximizing investment decisions and offset the negative impacts of high PR. The above arguments lead to the following hypothesis:

Hypothesis 4: Firms with high CSR have higher values when PR is high.

3. Data and Empirical Methodology

Our initial sample includes all the constituents of the Russell 3000 Index which is composed of 3000 large U.S. companies, as measured by market capitalization. The Russell 3000 Index represent 98% of the investable U.S. equity market. The list of constituents of the Russell 3000 Index between 2010 and 2020 is obtained from Bloomberg. To construct our final sample of firms, we start by removing financial and utility firms from the list of constituents of the Russell 3000 Index due to their different balance sheet structures and the government support they may receive. Then, we require that the remaining sample of firms have data in these three databases: Refinitiv ESG database for CSR data, Refinitiv DataStream for accounting data, and the PR data from [53]. Since PR data is measured on a quarterly basis, we use the data of the last quarter as a proxy for the yearly PR of a firm. Our final sample includes 16, 518 firm-year observations which correspond to 2055 unique firms over the sample period 2010–2020.

Ref. [53] developed a firm-level measure of PR that captures firm-level exposure to PR. It is constructed using information produced by decision makers within the firm rather than information produced outside the firm. More specifically, ref. [53] used a training library of political text and a training library of non-political text to identify two-word combinations (bigrams) that are often used in political texts [53]. Afterward, "[they] *count the number of instances in which these bigrams are used in a conference call in conjunction with synonyms for "risk" or "uncertainty" and divide by the total length of the call.*" [53] (pp.2136-2137) The result represents the share of the conversation related to PR [53]. We standardize this measure so that the distribution has zero mean and a standard deviation of one:

$$ZPRISK_{it} = \frac{PRISK_{it} - \overline{PRISK}}{SD_{PRISK}},\tag{1}$$

where PRISK is the measure of [53]. ZPRISK is the standardized measure of PR, \overline{PRISK} is the average of PR in the sample, and SD_{PRISK} is its sample standard deviation. For robustness, PRISK is only divided by SD_{PRISK} and used as a second standardized measure of PR.

Sustainability **2022**, 14, 11217 8 of 24

We obtain CSR data from the Refinitiv ESG database which collects ESG data from a variety of sources such as annual reports, CSR reports, company websites, NGO websites, stock exchange filings and news sources [76]. Refinitiv evaluates a firm's CSR based on three pillars: environmental, social, and corporate governance. Each pillar is given a score based on firm performance: Environment Pillar Score, Social Pillar Score and Governance Pillar Score. These scores include several categories. The environmental (E) pillar includes three categories: resource use, emissions, and innovation. The social (S) pillar includes four categories: workforce, human rights, community, and product responsibility. The governance (G) pillar includes three categories: management, shareholders, and corporate social responsibility strategy. Each category contains several ESG themes [76]. For example, the emission category includes four themes: CO2 emissions, waste, biodiversity, and environmental management systems. The workforce category includes four themes: diversity and inclusion, career development and training, working conditions, and health and safety. The shareholders category includes two themes: shareholder rights and takeover defenses. Refinitiv also calculates an overall ESG Score as the weighted average of the underlying 10 category scores where the category weights vary per industry for the environmental and social categories, i.e., based on relative performance and materiality, as the importance of ESG factors differ across industries [76]. All ESG scores range from 0 to 100.

Our dependant variable is Tobin's Q measured as the market value of total assets divided by book value of total assets. Tobin's Q is also measured as follows: (total assets – common equity + price close *common shares outstanding) /total assets. Our two main explanatory variables are CSR scores and ZPRISK. The latter is our proxy for firm-level PR. Following prior studies, we control for important firm characteristics (firm size, return on assets (ROA), leverage, cash, investment, Selling, General, and Administrative Expense (SGA), and sales growth). All variables, except CSR scores, are winsorized at the 1st and 99th percentiles to reduce the impact of outliers. Table 1 presents the variable definitions.

Table 1. Variables definitions.

| Variable | Measurement | | | |
|---------------------------|--|--|--|--|
| | Market Value of Total Assets divided by Book Value of Total Assets | | | |
| Tobin's Q | = (Total Assets – Common Equity + Price Close * Common Shares | | | |
| | Outstanding)/Total Assets | | | |
| Overall CSR Score | ESG Score | | | |
| E Score | Environment Pillar Score | | | |
| S Score | Social Pillar Score | | | |
| G Score | Governance Pillar Score | | | |
| ZPRISK | Standardized Measure of Political Risk Index From [23] | | | |
| Firm's Size | Natural Logarithm of Total Assets | | | |
| ROA | Net Income/Total Assets | | | |
| Lavaraga | (Long-Term Debt + Debt in Current Liabilities - Cash and Marketa | | | |
| Leverage | ble Securities) /Total Assets | | | |
| Cash | Cash and Short-Term Investments /Total Assets | | | |
| Investment | Capex / Total Assets | | | |
| Selling, General, and Ad- | SGA / Total Assets | | | |
| ministrative Expense | | | | |
| Sales Growth | Natural Logarithm of (sales at t / sales at t - 1) | | | |

To implement our empirical tests, we follow [34] and propose the following regression:

$$Q_{i,t} = \beta_0 + \beta_1 CSR_{i,t-1} + \beta_2 ZPRISK_{i,t-1} + \beta_3 CSR_{i,t-1} * ZPRISK_{i,t-1} + Controls_{i,t-1} + \epsilon_i$$
 (2)

where $Q_{i,t}$ is Tobin's Q, CSR is firm social scores, and ZPRISK is PR. In our baseline specifications, we examine the impact of an overall CSR score and three CSR sub-scores related to environment (E), social (S), and governance (G) issues. To mitigate endogeneity

Sustainability **2022**, 14, 11217 9 of 24

concerns we lag CSR, PR and control variables by one period. We also include industry-and year-fixed effects. We estimate our regressions using the Ordinary Least Square (OLS) method in the software STATA 17. The reported p-values of the estimated coefficients are based on the robust standard errors adjusted for clustering at the firm level. The estimated coefficients are robust to heteroscedasticity and autocorrelation of the error terms. For robustness, we also examine the impact of 8 different political topics in addition to the baseline measure of overall exposure to PR (Equation 1). Hypothesis 1 suggests that coefficient β_2 should be negative and significant. The same coefficient should be nonsignificant under hypothesis 2. Furthermore, a positive and significant β_1 provides support for hypothesis 3. Finally, under hypothesis 4, the coefficient of the interaction between CSR and ZPRISK (β_3) should be positive and significant.

4. Results

Descriptive statistics of the sample are presented in Table 2. We report the mean, median, standard deviation, minimum, maximum, 25th percentile, and 75th percentile of the main variables used in our analysis. The dependent variable is firm value proxied by Tobin's Q. The mean (median) Tobin's Q is 2.454 (1.791). Our first key independent variable is the standardized measure of PR (overall PR) and its eight subcomponents (economic, environment, trade, institutions, health, security, tax, and technology). We use eight different channels of PR in addition to an overall measure because some specific channels of PR may have stronger, weaker, or neutral valuation effects. On average, the standardized overall measure of PR is 0 with a standard deviation of 1. The minimum value is –0.537 and the maximum value is 22.443. The overall measure of CSR has a mean (median) value of 41.089 (37.76).

N Mean Std. Dev. p25 Median p75 min max Q 16,410 2.454 1.891 1.307 1.791 2.831 0.73 11.459 CSR score 11,821 41.089 19.282 26.04 54.66 37.76 0.57 95.19 11,821 43.857 21.218 27.41 40.77 58.57 0.46 98.12 S-score 98.55 11,821 26.87 27.897 E-score 0 17.13 48.77 0 49.048 22.044 31.32 0.25 99.45 G-score 11,821 49.68 66.75 **ZPRisk** 16,518 0 -0.451-0.2750.067 -0.53722.443 ZPRisk economic 16,518 0 -0.424-0.2720.033 -0.50321.118 1 ZPRisk environment 16,518 0 1 -0.312-0.216-0.004-0.35557.602 ZPRisk trade 16,518 0 1 -0.182-0.138-0.034-0.19595.116 ZPRisk institutions 16,518 0 1 -0.372-0.253-0.011-0.43219.825 ZPRisk health 16,518 0 1 -0.222-0.163-0.036-0.24843.321 0 ZPRisk security 16,518 1 -0.408-0.2620.023 -0.4822.936 -0.2310 38.995 ZPRisk tax 16,518 1 -0.337-0.009-0.3810 0.012 -0.439ZPRisk technology 16,518 1 -0.387-0.2640.863 14.446 1.714 14.452 SIZE 16,410 13.247 15.583 10.456 18.692 ROA 16,408 0.009 0.156 -0.0030.04 0.079 -0.7620.293LEVERAGE 15,849 0.067 0.363 -0.1550.115 0.319 -0.9060.91 0.197 0.218 0.038 0.002 0.933 CASH 16,352 0.113 0.271 INVESTMENT 16,397 0.045 0.048 0.015 0.029 0.056 0 0.269 SGA expense 16,055 0.262 0.239 0.089 0.196 0.363 0.005 1.21 SALES growth 13,959 0.07 0.23 -0.0090.061 0.148 -0.8550.994

Table 2. Summary statistics.

Note: This Table reports the summary statistics of the sample. The dependent variable is Tobin's Q. The key independent variables are CSR overall score, three CSR sub-scores (E, S, G), an overall standardized measure of PR proposed by [53], and eight subcomponents of PR measure (economic, environment, trade, institutions, health, security, tax, and technology). All variables are defined in Table 1.

Table 3 reports the estimated results of our baseline regressions using aggregate measure of CSR (overall CSR score), firm-level measure of PR (overall ZPRISK), and disaggregated measures of CSR (E, S, and G scores). In model (1), we can observe the effect

of overall CSR scores and overall ZPRISK on firm value. There is a positive and significant relationship between Tobin's Q and the overall CSR score (β_1 is 0.006 and significant at the 1% level). This result is consistent with hypothesis 3. However, there is no significant association between Tobin's Q and the overall ZPRISK, indicating a neutral impact of PR on firm value. The coefficient β_2 is not statistically significant. This result does not support hypothesis 1 which suggests a negative relation between PR and firm value. Instead, this result provides support to hypothesis 2 which suggests that adequate risk management strategies during high-PR periods can preserve firm value. The coefficient β_3 associated with the interaction term CSR*ZPRISK informs us about whether firms with strong (weak) CSR engagement have higher (lower) values when PR is higher (lower). The result suggests that PR has no effect on the CSR–firm value relationship. Our findings indicate that CSR adds value to firms (β_1 is positive and significant) but that there are no incremental benefits linked to CSR during times of high PR (β_3 is not significant).

Models (2) and (3) in Table 3 repeat the same analysis using S-score and E-score as proxies for firm CSR engagement. The coefficient β_1 is positive and significant in both models (2) and (3) suggesting that high scores in the S and E dimensions also increase firm value. In model (2) and (3), the coefficients β2 and β3 are not statistically significant suggesting again that the impact of PR on firm value and on the CSR-firm value relationship is neutral. In model (4), we use G-score to measure firm CSR engagement. The CSR-firm value association becomes negative and significant at the 5% level (β_1 is -0.002 with pvalue of 0.025) suggesting that firms with high G-scores have lower firm values. Consistent with our previous results, the impact of PR on firm value and on the CSR-firm value relationship remains neutral when using G-score to measure firm CSR engagement. Finally, we examine the three CSR dimensions simultaneously in model (5). It seems that the positive and significant relationship between CSR and firm value is driven by the Sscore. High G-score results in lower firm value. The coefficient associated with the E-score becomes insignificant. The coefficients β2 and β3 remain insignificant. Thus, the benefits of CSR are not dependent on whether PR is high or low. CSR engagement appears to create value primary during "normal" and more "steady-state" times. These benefits do not increase when PR is high. In all models of Table 3, the coefficients on control variables are consistent with prior literature. For instance, our findings indicate a significant and positive association between Tobin' Q and ROA, leverage, cash, investment, SGA expenses, and sales growth.

Table 3. Impact of PR on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5 |
|------------------|-----------|-----------|-----------|-----------|----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.131) |
| G score | | | | -0.002** | -0.003** |
| | | | | (0.025) | (0.000) |
| sPRisk | 0.005 | 0.011 | -0.015 | 0.003 | 0.028 |
| | (0.927) | (0.810) | (0.607) | (0.948) | (0.650) |
| sPRisk*CSR score | -0.001 | | | | |
| | (0.613) | | | | |
| sPRisk*S score | | -0.001 | | | -0.001 |
| | | (0.448) | | | (0.296) |
| sPRisk*E score | | | -0.000 | | 0.001 |
| | | | (0.951) | | (0.456) |
| SPRisk*G score | | | | -0.000 | -0.000 |
| | | | | (0.699) | (0.804) |
| sizew | -0.122*** | -0.134*** | -0.120*** | -0.063*** | -0.134** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.884*** | 3.902*** | 3.915*** | 3.964*** | 3.946*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.111*** | 1.111*** | 1.105*** | 1.064*** | 1.105*** |
| · · | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash stiw | 5.012*** | 4.967*** | 4.998*** | 4.995*** | 4.937*** |
| _ | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.867*** | 1.886*** | 1.844*** | 1.809*** | 1.918*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.387*** | 2.364*** | 2.419*** | 2.484*** | 2.377*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales gw | 0.975*** | 0.963*** | 0.969*** | 0.909*** | 0.945*** |
| _0 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.583*** | 1.734*** | 1.657*** | 0.892*** | 1.842*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Note: All models are estimated using industry- and year-fixed effects. *, **, and *** denote statistical significance at 10%, 5%, and 1%, respectively. All variables are defined in Table 1.

To further test the robustness of our results, we decompose PR into eight different types of PR and examine the impact of each component separately. More specifically, we examine the different topics of firm-level PR related to economic, environment, trade, institutions, health, security, tax, and technology issues. To validate their eight topic-specific measures of PR, [53] exploited firms' disclosure of the total expenditures on lobbying and the list of specific political topics these expenditures are directed toward. Using these disclosures, [23] created a map between topics discussed in conference calls and topics connected to firms' lobbying. The mapping shows that an increase of one standard deviation in risk related to a specific political topic is associated with an 11% increase in the probability that a given firm will lobby on that topic [53]. We find that PR stemming from economic (Table 4), environment (Table 5), health (Table 6), institutions (Table 7), tax (Table 8), and technology (Table 9) components have no effect on firm value and on the CSR-firm value relationship. Furthermore, we find weak evidence suggesting that the effect of

Sustainability **2022**, 14, 11217 12 of 24

PR is significant (at the 10% level) for the trade component (Table 10). Models (2) and (3) of Table 10 show that PR stemming from the topic of trade has a negative effect on firm value and a positive effect on the CSR–firm value relationship when using S and E scores. We re-estimate our regressions using security as a specific topic for PR in Table 11. The moderating effect of PR on the CSR–firm value relationship is negative and significant (at the 5% level) when using S-score to measure CSR engagement. In sum, our additional tests show that firm-level PR is not a moderator of the CSR–firm value relationship, regardless of the PR type.

Table 4. Impact of PR stemming from economic topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5 |
|----------------------------|-----------|-----------|-----------|-----------|----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.137) |
| G score | | | | -0.002** | -0.003** |
| | | | | (0.024) | (0.000) |
| sPRiskT_economic | 0.046 | 0.040 | 0.011 | 0.022 | 0.053 |
| | (0.383) | (0.419) | (0.778) | (0.599) | (0.277) |
| sPRiskT_economic*CSR score | -0.001 | | | | |
| _ | (0.222) | | | | |
| sPRiskT_economic*S score | • • | -0.001 | | | -0.001 |
| _ | | (0.231) | | | (0.218) |
| sPRiskT economic*E score | | , , | -0.001 | | 0.000 |
| _ | | | (0.547) | | (0.862) |
| sPRiskT economic*G score | | | ` ′ | -0.000 | -0.000 |
| _ | | | | (0.508) | (0.733) |
| sizew | -0.122*** | -0.134*** | -0.120*** | -0.063*** | -0.134** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.883*** | 3.901*** | 3.916*** | 3.964*** | 3.946** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.112*** | 1.112*** | 1.106*** | 1.064*** | 1.105** |
| 8 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash stiw | 5.010*** | 4.963*** | 4.997*** | 4.994*** | 4.932*** |
| _ | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.881*** | 1.900*** | 1.859*** | 1.820*** | 1.933*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.389*** | 2.366*** | 2.422*** | 2.486*** | 2.379** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales gw | 0.975*** | 0.963*** | 0.968*** | 0.908*** | 0.944** |
| 228 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.586*** | 1.738*** | 1.655*** | 0.892*** | 1.844** |
| Constant | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| | (3.3.3.) | () | (*****) | (*****) | (5.550) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 5. Impact of PR stemming from environmental topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|-------------------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.007*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.151) |
| G score | | | | -0.002** | -0.003*** |
| | | | | (0.025) | (0.000) |
| sPRiskT environment | 0.039 | 0.029 | 0.028 | 0.017 | 0.014 |
| _ | (0.477) | (0.582) | (0.447) | (0.770) | (0.820) |
| sPRiskT environment*CSR score | -0.001 | , , | , , | ` , | ` |
| _ | (0.458) | | | | |
| sPRiskT environment*S score | , , | -0.001 | | | 0.000 |
| _ | | (0.565) | | | (0.879) |
| sPRiskT environment*E score | | , | -0.001 | | -0.001 |
| _ | | | (0.276) | | (0.289) |
| sPRiskT environment*G score | | | , | -0.000 | 0.000 |
| _ | | | | (0.855) | (0.872) |
| sizew | -0.122*** | -0.133*** | -0.120*** | -0.063*** | -0.133*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.883*** | 3.902*** | 3.914*** | 3.964*** | 3.946*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.112*** | 1.112*** | 1.106*** | 1.064*** | 1.104*** |
| 5 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash stiw | 5.009*** | 4.964*** | 4.995*** | 4.992*** | 4.934*** |
| _ | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.883*** | 1.901*** | 1.863*** | 1.827*** | 1.938*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.391*** | 2.367*** | 2.424*** | 2.487*** | 2.381*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales gw | 0.975*** | 0.964*** | 0.968*** | 0.908*** | 0.944*** |
| _6 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.582*** | 1.731*** | 1.655*** | 0.892*** | 1.832*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 6. Impact of PR stemming from health topic on the CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|--------------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.152) |
| G score | | | | -0.002** | -0.003*** |
| | | | | (0.025) | (0.000) |
| sPRiskT_health | 0.064 | 0.056 | 0.018 | 0.019 | 0.053 |
| | (0.230) | (0.494) | (0.386) | (0.673) | (0.538) |
| sPRiskT_health*CSR score | -0.002 | | | | |
| | (0.200) | | | | |
| sPRiskT_health*S score | | -0.001 | | | -0.001 |
| | | (0.451) | | | (0.611) |
| sPRiskT_health*E score | | | -0.001 | | -0.001 |
| | | | (0.254) | | (0.604) |
| sPRiskT_health*G score | | | | -0.000 | -0.000 |
| | | | | (0.750) | (0.977) |
| sizew | -0.122*** | -0.133*** | -0.119*** | -0.063*** | -0.133*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.888*** | 3.905*** | 3.918*** | 3.965*** | 3.950*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.112*** | 1.113*** | 1.107*** | 1.064*** | 1.106*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash_stiw | 5.009*** | 4.965*** | 4.997*** | 4.993*** | 4.934*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.878*** | 1.898*** | 1.856*** | 1.824*** | 1.930*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.389*** | 2.367*** | 2.422*** | 2.486*** | 2.380*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales_gw | 0.973*** | 0.962*** | 0.967*** | 0.908*** | 0.943*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.583*** | 1.735*** | 1.653*** | 0.892*** | 1.836*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 7. Impact of PR stemming from institutions topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|--------------------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | 0.000 | (0.129) |
| G score | | | | -0.002** | -0.003*** |
| DD' 1 Th. ' ' ' | 0.010 | 0.012 | 0.014 | (0.024) | (0.000) |
| sPRiskT_institutions | 0.019 | 0.012 | -0.014 | 0.025 | 0.046 |
| DD: 1.T. ' .'' *CGD | (0.668) | (0.759) | (0.592) | (0.602) | (0.370) |
| sPRiskT_institutions*CSR score | -0.001 | | | | |
| sPRiskT institutions*S score | (0.362) | -0.001 | | | -0.001 |
| SFRISKI_HISHILLIOHS S SCOLE | | (0.406) | | | (0.366) |
| sPRiskT_institutions*E score | | (0.400) | -0.000 | | 0.001 |
| SI KISKI_IIISHILIIOIIS L SCOIC | | | (0.899) | | (0.343) |
| sPRiskT institutions*G score | | | (0.055) | -0.001 | -0.001 |
| si Riski_msitations o score | | | | (0.364) | (0.350) |
| sizew | -0.122*** | -0.133*** | -0.120*** | -0.063*** | -0.134*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.885*** | 3.902*** | 3.916*** | 3.966*** | 3.947*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.112*** | 1.111*** | 1.106*** | 1.065*** | 1.105*** |
| - | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash_stiw | 5.012*** | 4.966*** | 4.999*** | 4.996*** | 4.935*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.868*** | 1.888*** | 1.845*** | 1.809*** | 1.919*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.388*** | 2.365*** | 2.420*** | 2.485*** | 2.379*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales_gw | 0.974*** | 0.963*** | 0.969*** | 0.908*** | 0.945*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.583*** | 1.732*** | 1.654*** | 0.892*** | 1.841*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 8. Impact of PR stemming from tax topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|-----------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.148) |
| G score | | | | -0.002** | -0.003*** |
| | | | | (0.026) | (0.000) |
| sPRiskT_tax | 0.013 | 0.022 | 0.016 | -0.015 | -0.014 |
| _ | (0.779) | (0.601) | (0.669) | (0.765) | (0.791) |
| sPRiskT tax*CSR score | -0.000 | ` / | ` ′ | ` , | ` ′ |
| _ | (0.792) | | | | |
| sPRiskT tax*S score | ` / | -0.000 | | | -0.000 |
| _ | | (0.600) | | | (0.839) |
| sPRiskT_tax*E score | | () | -0.000 | | -0.001 |
| | | | (0.607) | | (0.633) |
| sPRiskT_tax*G score | | | (*****) | 0.000 | 0.001 |
| 57 Tubir 1_0mr | | | | (0.648) | (0.404) |
| sizew | -0.122*** | -0.133*** | -0.120*** | -0.063*** | -0.133*** |
| 5.25 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.885*** | 3.903*** | 3.915*** | 3.964*** | 3.946*** |
| 19411 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.111*** | 1.111*** | 1.106*** | 1.064*** | 1.104*** |
| ie verage w | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash stiw | 5.010*** | 4.966*** | 4.997*** | 4.993*** | 4.936*** |
| cusii_suw | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.885*** | 1.905*** | 1.866*** | 1.829*** | 1.947*** |
| m v estiment vv | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.390*** | 2.366*** | 2.423*** | 2.487*** | 2.380*** |
| Costw | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales_gw | 0.975*** | 0.964*** | 0.969*** | 0.908*** | 0.944*** |
| saics_gw | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.579*** | 1.729*** | 1.653*** | 0.890*** | 1.827*** |
| Collstant | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Sustainability **2022**, 14, 11217 17 of 24

Table 9. Impact of PR stemming from technology topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|------------------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.007*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.136) |
| G score | | | | -0.002** | -0.003*** |
| | | | | (0.024) | (0.000) |
| sPRiskT_technology | -0.012 | -0.017 | -0.017 | 0.021 | 0.009 |
| | (0.729) | (0.584) | (0.460) | (0.511) | (0.763) |
| sPRiskT technology*CSR score | 0.000 | | | , , | , , |
| _ | (0.690) | | | | |
| sPRiskT technology*S score | ` ′ | 0.000 | | | 0.000 |
| _ 27 | | (0.532) | | | (0.681) |
| sPRiskT technology*E score | | , | 0.001 | | 0.001 |
| _ 23 | | | (0.251) | | (0.209) |
| sPRiskT technology*G score | | | , | -0.000 | -0.001 |
| _ 87 | | | | (0.501) | (0.256) |
| sizew | -0.122*** | -0.133*** | -0.120*** | -0.063*** | -0.133*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.886*** | 3.905*** | 3.917*** | 3.965*** | 3.952*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.111*** | 1.112*** | 1.107*** | 1.064*** | 1.107*** |
| 22 / 22.182 // | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash stiw | 5.011*** | 4.967*** | 4.996*** | 4.993*** | 4.936*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.880*** | 1.896*** | 1.851*** | 1.820*** | 1.916*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.390*** | 2.367*** | 2.423*** | 2.486*** | 2.380*** |
| 3 555 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales gw | 0.975*** | 0.964*** | 0.970*** | 0.908*** | 0.946*** |
| 222 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.578*** | 1.725*** | 1.653*** | 0.892*** | 1.834*** |
| 00110111111 | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 10. Impact of PR stemming from trade topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|-------------------------|-----------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** | | | | |
| | (0.000) | | | | |
| S score | | 0.007*** | | | 0.006*** |
| | | (0.000) | | | (0.000) |
| E score | | | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.139) |
| G score | | | | -0.002** | -0.003*** |
| | | | | (0.026) | (0.000) |
| sPRiskT_trade | -0.062 | -0.060* | -0.054* | 0.035 | 0.009 |
| | (0.160) | (0.073) | (0.087) | (0.499) | (0.869) |
| sPRiskT_trade*CSR score | 0.001 | | | | |
| | (0.168) | | | | |
| sPRiskT_trade*S score | | 0.001* | | | 0.000 |
| | | (0.052) | | | (0.924) |
| sPRiskT_trade*E score | | | 0.001* | | 0.001 |
| | | | (0.061) | | (0.543) |
| sPRiskT_trade*G score | | | | -0.001 | -0.001 |
| | | | | (0.339) | (0.143) |
| sizew | -0.122*** | -0.133*** | -0.120*** | -0.063*** | -0.133*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.889*** | 3.908*** | 3.921*** | 3.964*** | 3.951*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.110*** | 1.110*** | 1.105*** | 1.064*** | 1.105*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash_stiw | 5.013*** | 4.971*** | 5.000*** | 4.995*** | 4.942*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.870*** | 1.886*** | 1.844*** | 1.820*** | 1.916*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.385*** | 2.362*** | 2.417*** | 2.487*** | 2.377*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales_gw | 0.975*** | 0.964*** | 0.969*** | 0.908*** | 0.945*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.578*** | 1.725*** | 1.656*** | 0.886*** | 1.831*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Table 11. Impact of PR stemming from security topic on CSR-Q link.

| VARIABLES | Model (1) | Model (2) | Model (3) | Model (4) | Model (5) |
|---|---------------------|-----------|-----------|-----------|-----------|
| CSR score | 0.006*** (0.000) | | | | |
| S score | (0.000) | 0.007*** | | | 0.006*** |
| 2 | | (0.000) | | | (0.000) |
| E score | | (*****) | 0.004*** | | 0.001 |
| | | | (0.000) | | (0.137) |
| G score | | | , , | -0.002** | -0.003*** |
| | | | | (0.026) | (0.000) |
| sPRiskT_security | 0.046 | 0.061 | 0.009 | -0.000 | 0.046 |
| | (0.255) | (0.127) | (0.743) | (0.999) | (0.249) |
| sPRiskT security*CSR score | -0.001 | , , | ` , | , , | · ´ |
| | (0.196) | | | | |
| sPRiskT_security*S score | | -0.001* | | | -0.002** |
| | | (0.055) | | | (0.011) |
| sPRiskT_security*E score | | | -0.000 | | 0.001 |
| | | | (0.753) | | (0.283) |
| sPRiskT_security*G score | | | | 0.000 | 0.001 |
| | | | | (0.905) | (0.402) |
| sizew | -0.122*** | -0.134*** | -0.120*** | -0.063*** | -0.134*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| roaw | 3.883*** | 3.898*** | 3.916*** | 3.965*** | 3.942*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| leveragew | 1.111*** | 1.110*** | 1.106*** | 1.064*** | 1.103*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| cash_stiw | 5.008*** | 4.960*** | 4.996*** | 4.993*** | 4.928*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| investmentw | 1.883*** | 1.901*** | 1.862*** | 1.827*** | 1.939*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| costw | 2.390*** | 2.365*** | 2.423*** | 2.487*** | 2.379*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| sales_gw | 0.974*** | 0.962*** | 0.969*** | 0.908*** | 0.944*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Constant | 1.582*** | 1.738*** | 1.654*** | 0.891*** | 1.846*** |
| | (0.000) | (0.000) | (0.000) | (0.000) | (0.000) |
| Observations | 8559 | 8559 | 8559 | 8559 | 8559 |
| R-squared | 0.379 | 0.381 | 0.379 | 0.378 | 0.382 |
| Industry FE | Yes | Yes | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes | Yes | Yes |

Note: All models are estimated with industry- and year-fixed effects. *, **, and *** denote statistical significance at 10%, 5%, and 1%, respectively. All variables are defined in Table 1.

Finally, we re-estimate Equation (2) for each industry in our sample (we have nine industries: communications, consumer discretionary, consumer staples, energy, health care, industrials, materials, real estate, and technology). This allows us to better control industry effects and make sure they are not driving our findings. The results suggest that the neutral impact of PR on firm value and on the CSR–firm value relationship holds for all industries, suggesting a uniform neutral impact across industries. For the sake of brevity, we do not report these results.

5. Discussion

This paper examines whether firm-level (idiosyncratic) political risk (PR) affects the relationship between corporate social responsibility (CSR) and firm value. We developed four hypotheses. The first and second hypotheses are mutually exclusive; the first hypothesis predicts a negative relationship between PR and firm value, while the second

Sustainability **2022**, 14, 11217 20 of 24

hypothesis predicts a neutral relationship between PR and firm value. The third hypothesis suggests that CSR is positively related to firm value, whereas the fourth hypothesis suggests that PR will moderate the CSR–firm value relationship. Our main findings are as follows. First, firm-level PR does not affect firm value. Second, CSR is positively related to firm value. Finally, PR has no effect on the CSR–firm value relationship. In sum, our results provide support only to the second and third hypotheses.

Our first finding indicates that, on average, variations in firm-specific PR do not explain differences in firm value. Most studies in the literature rely on the real option channel and informational channel to explain the valuation effects of high PR. Our first finding provides support to alternative channel which is the managerial risk-taking incentives. The option-value of an investment delay can be high when PR is high [36,45]. PR can also increase managerial risk aversion. The board of directors understand that high PR coupled with managers' inherent risk aversion (loss aversion) can push firms' managers to choose the option-to-wait component of investment decisions [43,63]. This can result in suboptimal risk-taking strategies when managers choose not to support high-risk and high-yield investment opportunities [43,63]. The managerial risk-taking incentives channel assumes that PR carries not only the possibility for delayed investments but also the possibility for new potential investment opportunities created by a high-PR environment. It also assumes that managers can avoid "negative" uncertainty while making use of " positive" uncertainty during high-PR periods [43,62,63]. In this respect, boards can provide more powerful and optimal risk-taking incentives that allow managers to support adequately high-risk and high-yield investment projects in periods of severe PR. A direct managerial implication of this is that well-designed risk-taking incentives during times of high PR can help offset the negative effects of PR, if any. That is, adequate risk-management strategies during high-PR periods can preserve firm value.

Our second finding suggests a positive relationship between CSR and firm value. This is consistent with the results reported in previous studies [12–17] as well as the stakeholder theory which predicts that CSR adds value to a firm. This positive relationship between CSR and firm value is mainly driven by the social component of CSR. However, the governance component of CSR is negatively related to firm value, whereas the environment component of CSR has no impact on firm value. Therefore, the different CSR components have different impacts on firm value. This result highlights the importance of examining the individual CSR dimensions along with the aggregate CSR and has implications for academic research, managers, investors, and policymakers. For instance, several studies [27,33,34] consider only E and S when computing their aggregate CSR measure. They argue that governance is not part of a firm's CSR remit. However, ESG data providers such as Refinitiv, Bloomberg, and S&P Global Market Intelligence as well as investors (as evidenced by the UN backed PRI) consider G as part of a firm's CSR engagement. We argue that this gap between academics and practitioners is an important issue that could be addressed in future research. In [27], the researchers argued that the link between CSR and corporate governance is unexplored and called for further work on this topic.

Our third finding suggests that firm-level PR is not a moderator of the CSR-firm value relationship, regardless of the PR type. Thus, the benefits of CSR are not affected by the level of firm-specific (idiosyncratic) exposure to PR. The authors of [34] found that CSR is positively related to firm value only when the economic policy uncertainty index (a proxy for systematic political risk) is high. In a sense, our findings complement those reported by [34]. In ref. [34], the authors show that systematic PR is priced in the financial market so that it moderates the CSR-firm value relationship. Our finding suggests that firm-level (idiosyncratic) political risk (PR) is not priced in the financial market and as such it does not affect the CSR-firm value relationship. Neither firm-specific PR in aggregate nor the individual PR components affect the relationship between CSR and firm value. It is important to note that idiosyncratic PR is only one component of the firm's overall idiosyncratic risk. Indeed, portfolio theory suggests that only systematic risk is

Sustainability **2022**, 14, 11217 21 of 24

priced. Firm-specific risk is irrelevant for investors as this could be diversified away when constructing well-diversified portfolios such as the Russell 3000 Index considered in this study.

In addition, CSR engagement appears to create value during "normal" and more "steady-state" times. There are neither additional benefits nor penalties when PR is high. This should encourage managers and increase their incentives to pursue CSR activities without worrying much about their firm-specific PR. It is important to note that this is true for firms with specific characteristics such as those included in the Russell 3000 Index, e.g., larger firms. This should also inform policymakers when developing and implementing their policies designed to promote CSR activities. For instance, policymakers could be inspired from the best CSR practices of industry CSR leaders, e.g., those included in the Russell 3000 Index, and focus their sustainability efforts (e.g., conversion to low-carbon economy, more CSR disclosure and transparency) toward medium and small firms. Policymakers could help them achieve this by providing tax incentives, better access to certain markets, or better access to financing (e.g., some forms of guaranties).

6. Conclusions

The aim of this paper is to investigate whether firm-level (idiosyncratic) political risk (PR) affects the relationship between corporate social responsibility (CSR) and firm value. Based on real option theory and informational argument (managerial risk-taking incentives), we hypothesize that firm-level PR will have a negative (neutral) impact on firm value. Based on the stakeholder theory, we hypothesize that CSR will have a positive impact on firm value. We also hypothesize that PR will moderate the CSR–firm value relationship. We test these hypotheses using a sample of 16,518 firm-year observations which correspond to 2055 unique firms belonging to the Russell 3000 Index over the sample period 2010–2020.

Our main findings are as follows. First, firm-level PR does not affect firm value suggesting that adequate risk-management strategies during high-PR periods can preserve firm value. Second, CSR is positively related to firm value which is mainly driven by the social component of CSR. However, the governance component of CSR is negatively related to firm value, whereas the environment component of CSR has no impact on firm value. Therefore, the different CSR components have different impacts on firm value. Finally, PR has no effect on the CSR-firm value relationship, regardless of the PR type. Thus, the benefits of CSR are not affected by the level of firm-specific (idiosyncratic) exposure to PR. Our finding suggests that firm-level PR is not priced in the financial market and as such it does not affect the CSR-firm value relationship. This is consistent with portfolio theory which suggests that only systematic risk is priced.

Our work has some limitations that are left for future research. First, our sample selection criteria are restricted to firms belonging to the Russell 3000 Index. This implies that our results could not be generalized to all type of firms and across countries. Therefore, future research could examine the impact of firm-specific PR on the CSR-firm value relationship for other firms not included in such indices (e.g., small and medium firms) in developed and developing countries. For instance, ref. [35] found that politically connected Chinese firms overinvest in CSR to obtain political legitimacy and access to governmental resources. Political connection is perceived negatively by investors. Second, our findings regarding the positive impact of CSR on firm value could be affected by potential endogeneity issues, e.g., omitted factors that affect both CSR and firm value or simultaneity bias between both variables. Addressing this endogeneity issue is challenging because it requires identifying instruments correlated with CSR, but not with firm value or the use of an exogenous shock affecting CSR (e.g., regulatory change). Finally, our work could be seen as exploratory and leads to several interesting research topics. For example, future research could further examine the effects of the different firm-specific PR components (economic, environment, trade, institutions, health, security, tax, and technology) on the different sub-components of the E, S and G dimensions (e.g., diversity, Sustainability **2022**, 14, 11217 22 of 24

community, and human rights within the S; climate risk within the E dimension; board independence within the G dimension) as some specific channels of PR may have stronger, weaker, or neutral valuation effects. It would be fruitful to also explore industry variation as stakeholders' expectations and CSR issues vary across industries.

Author Contributions: Conceptualization, A.H.; methodology, A.H. and K.B.; software, A.H.; validation, A.M. and A.H.; formal analysis, A.M. and A.H.; investigation, A.M. and A.H.; data curation, K.B.; writing—original draft preparation, A.M.; writing—review and editing, A.M., A.H. and K.B.; visualization, A.M.; supervision, A.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The ESG and financial data that support the findings of this study are available from Refinitiv. Restrictions apply to the availability of these data, which were used under license for this study, and so are not publicly available. Political risk data are publicly available and can be downloaded at: https://www.firmlevelrisk.com/home.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Hale, Z. Biden Signs Historic Climate Bill, Teeing Up Surge in US Clean Energy Spending; S&P Global Market Intelligence: New York, NY, USA, August 16 2022.
- 2. Friedman, M. The social responsibility of business is to increase its profits. *The New York Times Magazine*, 13 September 1970, pp. 32–33, 122, 124, 126.
- 3. Cespa, G.; Cestone, G. Corporate Social Responsibility and Managerial Entrenchment. *J. Econ. Manag. Strat.* **2007**, *16*, 741–771. https://doi.org/10.1111/j.1530-9134.2007.00156.x.
- 4. Surroca, J.; Tribó, J.A. Managerial Entrenchment and Corporate Social Performance. J. Bus. Financ. Account. 2008, 35, 748–789. https://doi.org/10.1111/j.1468-5957.2008.02090.x.
- 5. Barnea, A.; Rubin, A. Corporate social responsibility as a conflict between shareholders. *J. Bus. Ethics* **2010**, *97*, 71–86. https://doi.org/10.1007/s10551-010-0496-z.
- 6. Freeman, R.E. Strategic Management: A Stakeholder Approach, 1st ed.; Pitman: Marshfield, WI, USA, 1984.
- Jones, T.M. Instrumental stakeholder theory: A synthesis of ethics and economics. Acad. Manag. Rev. 1995, 20, 404–437. https://doi.org/10.2307/258852.
- 8. Donaldson, T.; Preston, L. The stakeholder theory of the modern corporation: Concepts, evidence and implications. *Acad. Manag. Rev.* **1995**, *20*, 65–91. https://doi.org/10.2307/258887.
- 9. Waddock, S.A.; Graves, S.B. The corporate social performance financial performance link. *Strateg. Manag. J.* **1997**, *18*, 303–319. https://www.jstor.org/stable/3088143.
- 10. Margolis, J.D.; Walsh, J.P. People and Profits: The Search for a Link between a Company's Social and Financial Performance, 1st ed.; Lawrence Erlbaum Associates: Mahwah, NJ, USA, 2001.
- 11. Lev, B.; Petrovits, C.; Radhakrishnan, S. Is doing good good for you? How corporate charitable contributions enhance revenue growth. *Strateg. Manag. J.* **2010**, *31*, 182–200. https://doi.org/10.1002/smj.810.
- 12. Orlitzky, M.; Schmidt, F.L.; Rynes, S.L. Corporate social and financial performance: A meta-analysis. *Organ. Stud.* **2003**, 24, 403–441. https://doi.org/10.1177/0170840603024003910.
- 13. Lin, C.-H.; Yang, H.-L.; Liou, D.-Y. The impact of corporate social responsibility on financial performance: Evidence from business in Taiwan. *Technol. Soc.* **2009**, *31*, 56–63. https://doi.org/10.1016/j.techsoc.2008.10.004.
- 14. Okamoto, D. Social relationship of a firm and the CSP–CFP relationship in Japan: Using artificial neural networks. *J. Bus. Ethics* **2009**, *87*, 117–132. https://doi.org/10.1007/s10551-008-9874-1.
- 15. Mishra, S.; Suar, D. Does corporate social responsibility influence firm performance of Indian companies? *J. Bus. Ethics* **2010**, *95*, 571–601. https://doi.org/10.1007/s10551-010-0441-1.
- 16. Oeyono, J.; Samy, M.; Bampton, R. An examination of corporate social responsibility and financial performance: A study of the top 50 Indonesian listed corporations. *J. Glob. Responsib.* **2011**, 2, 100–112. https://doi.org/10.1108/20412561111128555.
- 17. Zhang, F.; Jung, J.-Y. Changes in the Influence of Social Responsibility Activities on Corporate Value over 10 Years in China. *Sustainability* **2020**, *12*, 9506. https://doi.org/10.3390/su12229506.
- 18. Aupperle, K.E.; Carroll, A.B.; Hatfield, J.D. An empirical examination of the relationship between corporate social responsibility and profitability. *Acad. Manag. Rev.* **1985**, *28*, 446–463. https://doi.org/10.5465/256210.

Sustainability **2022**, 14, 11217 23 of 24

19. McWilliams, A.; Siegel, D. Corporate social responsibility and financial performance: Correlation or misspecification? *Strateg. Manag. J.* 2000, 21, 603–609. https://doi.org/10.1002/(SICI)1097-0266(200005)21:5<603::AID-SMJ101>3.0.CO;2-3.

- Eom, K.; Nam, G. Effect of Entry into Socially Responsible Investment Index on Cost of Equity and Firm Value. Sustainability 2017, 9, 717. https://doi.org/10.3390/su9050717.
- 21. Lys, T.; Naughton, J.P.; Wang, C. Signaling through corporate accountability reporting. *J. Account. Econ.* **2015**, *60*, 56–72. https://doi.org/10.1016/j.jacceco.2015.03.001.
- Singh, P.J.; Sethuraman, K.; Lam, J.Y. Impact of Corporate Social Responsibility Dimensions on Firm Value: Some Evidence from Hong Kong and China. Sustainability 2017, 9, 1532. https://doi.org/10.3390/su9091532.
- 23. Dhaliwal, D.S.; Li, O.Z.; Tsang, A.; Yang, Y.G. Voluntary nonfinancial disclosure and the cost of equity capital: The initiation of corporate social responsibility reporting. *Account. Rev.* **2011**, *86*, 59–100. https://www.jstor.org/stable/29780225.
- 24. Kim, Y.; Park, M.S.; Wier, B. Is earnings quality associated with corporate social responsibility? *Account. Rev.* **2012**, *87*, 761–796. https://www.jstor.org/stable/23245629.
- 25. Cheng, B.; Ioannou, I.; Serafeim, G. Corporate social responsibility and access to finance. *Strateg. Manag. J.* **2014**, *35*, 1–23. https://doi.org/10.1002/smj.2131.
- Kim, Y.; Li, H.; Li, S. Corporate social responsibility and stock price crash risk. J. Bank. Financ. 2014, 43, 1–13. https://doi.org/10.1016/j.jbankfin.2014.02.013.
- 27. Servaes, H.; Tamayo, A. The impact of corporate social responsibility on firm value: The role of customer awareness. *Manag. Sci.* **2013**, *59*, 1045–1061. https://doi.org/10.1287/mnsc.1120.1630.
- 28. Buchanan, B.; Cao, C.X.; Chen, C. Corporate social responsibility, firm value, and influential institutional ownership. *J. Corp. Financ.* 2018, 52, 73–95. https://doi.org/10.1016/j.jcorpfin.2018.07.004.
- 29. D'Amato, A.; Falivena, C. Corporate social responsibility and firm value: Do firm size and age matter? Empirical evidence from European listed companies. *Corp. Soc. Responsib. Environ.* **2019**, *27*, 909–924. https://doi.org/10.1002/csr.1855.
- 30. Maury, B. Strategic CSR and firm performance: The role of prospector and growth strategies. *J. Econ. Bus.* **2022**, *118*, 106031. https://doi.org/10.1016/j.jeconbus.2021.106031.
- 31. Kim, W.S.; Park, K.; Lee, S.H. Corporate Social Responsibility, Ownership Structure, and Firm Value: Evidence from Korea. *Sustainability* **2018**, *10*, 2497. https://doi.org/10.3390/su10072497.
- 32. Elif, A.S. Corporate social responsibility and financial performance: The moderating role of ownership concentration in Turkey. *Sustainability* **2019**, *11*, 3643. https://doi.org/10.3390/su11133643.
- 33. Lins, K.; Servaes, H.; Tamayo, A. Social capital trust, and firm performance: The value of corporate social responsibility during the financial crisis. *J. Financ.* **2017**, *72*, 1785–1824. https://doi.org/10.1111/jofi.12505.
- 34. Borghesi, R.; Chang, K.; Li, Y. Firm value in commonly uncertain times: The divergent effects of corporate governance and CSR. *Appl. Econ.* **2019**, *51*, 4726–4741. https://doi.org/10.1080/00036846.2019.1597255.
- 35. Xiong, C.; Zhang, K.; Zhao, X. Do Political Ties Cause Over-Investment in Corporate Social Responsibility? Empirical Evidence from Chinese Private Firms. *Sustainability* **2020**, *12*, 7203. https://doi.org/10.3390/su12177203.
- 36. Gulen, H.; Ion, M. Policy uncertainty and corporate investment. *Rev. Financ. Stud.* **2016**, 29, 523–564. https://doi.org/10.1093/rfs/hhv050.
- 37. Liu, J.; Wang, H. Economic policy uncertainty and the cost of capital. *Int. Rev. Financ. Anal.* 2022, 81, 102070. https://doi.org/10.1016/j.irfa.2022.102070.
- 38. Boutchkova, M.; Doshi, H.; Durnev, A.; Molchanov, A. Precarious Politics and Return Volatility. *Rev. Financ. Stud.* 2012, 25, 1111–1154. https://doi.org/10.1093/rfs/hhr100.
- Chahine, S.; Daher, M.; Saade, S. Doing good in periods of high uncertainty: Economic policy uncertainty, corporate social responsibility, and analyst forecast error. J. Financ. Stab. 2021, 56, 100919. https://doi.org/10.1016/j.jfs.2021.100919.
- 40. Nagar, V.; Schoenfeld, J.; Wellman, L. The effect of economic policy uncertainty on investor information asymmetry and management disclosures. *J. Account. Econ.* **2019**, *67*, 36–57. https://doi.org/10.1016/j.jacceco.2018.08.011.
- 41. Luo, Y.; Zhang, C. Economic policy uncertainty and stock price crash risk. Res. Int. Bus. Financ. 2020, 51, 101112. https://doi.org/10.1016/j.ribaf.2019.101112.
- 42. Yung, K.; Root, A. Policy uncertainty and earnings management: International evidence. *J. Bus. Res.* **2019**, *100*, 255–267. https://doi.org/10.1016/j.jbusres.2019.03.058.
- 43. Chatjuthamard, P.; Wongboosin, P.; Kongsompong, K.; Jiraporn, P. Does economic policy uncertainty influence executive risk-taking incentives? *Financ. Res.* **2020**, *37*, 101385. https://doi.org/10.1016/j.frl.2019.101385.
- 44. Pastor, L.; Veronesi, P. Political uncertainty and risk premia. J. Financ. Econ 2013, 110, 520–545. https://doi.org/10.1016/j.jfineco.2013.08.007.
- 45. Bonaime, A.; Gulen, H.; Ion, M. Does policy uncertainty affect mergers and acquisitions? *J. Financ. Econ* **2018**, 129, 531–558. https://doi.org/10.1016/j.jfineco.2018.05.007.
- 46. Colak, G.; Durnev, A.; Qian, Y. Political uncertainty and IPO activity: Evidence form U.S. gubernational elections. *J. Financ. Quant. Anal.* **2017**, *52*, 2523–2564. https://doi.org/10.1017/S0022109017000862.
- Qasem, A.; Wan-Hussin, W.N.; Ghaleb, B.A.A.; Bamahros, H.M. Institutional investors, political connections and stock recommendations from sell-side analysts: The case of Malaysia. J. Account. Emerg. Econ. 2022. https://doi.org/10.1108/JAEE-05-2021-0177.

Sustainability **2022**, 14, 11217 24 of 24

48. Guedhami, O.; Mansi, S.; Reeb, D.; Yasuda, Y. Economic policy and allocative distortions. J. Financ. Stab 2021, 56, 100923. https://doi.org/10.1016/j.jfs.2021.100923.

- 49. Baker, S.; Bloom, N.; Davis, S. Measuring Economic Policy Uncertainty. *Q. J. Econ.* **2016**, 131, 1593–1636. https://doi.org/10.1093/qje/qjw024.
- Dhaliwal, D.S.; Radhakrishnan, S.; Tsang, A.; Yang, Y.G. Nonfinancial disclosure and analyst forecast accuracy: International evidence on corporate social responsibility disclosure. Account. Rev. 2012, 87, 723–759. https://doi.org/10.2308/accr-10218.
- 51. El Ghoul, S.; Guedhami, O.; Kwook, C.; Mishra, D. Does corporate social responsibility affect cost of capital? *J. Bank. Financ.* **2011**, *35*, 2388–2406. https://doi.org/10.1016/j.jbankfin.2011.02.007.
- 52. Bouslah, K.; Kryzanowski, L.; M'Zali, B. The impact of the dimensions of social performance on firm risk. *J. Bank. Financ.* **2013**, 37, 1258–1273. https://doi.org/10.1016/j.jbankfin.2012.12.004.
- 53. Hassan, A.T.; Hollander, S.; van Lent, L.; Tahoun, A. Firm-level political risk: Measurement and effects. Q. J. Econ. 2019, 134, 2135–2202. https://doi.org/10.1093/qje/qjz021.
- 54. Brogaard, J.; Detzel, A. The assets-pricing implications of government economic policy uncertainty. *Manag. Sci.* **2015**, *61*, 3–18. https://doi.org/10.1287/mnsc.2014.2044.
- 55. Drobetz, W.; El Ghoul, S.; Guedhami, O.; Janzen, M. Policy uncertainty, investment, and the cost of capital. *J. Financ. Stab.* **2018**, 39, 28–45. https://doi.org/10.1016/j.jfs.2018.08.005.
- 56. Pan, W.-F.; Wang, X.; Yang, S. Debt maturity, leverage, and political uncertainty. N. Am. J. Econ. Financ. 2019, 50, 100981. https://doi.org/10.1016/j.najef.2019.04.024.
- 57. Dai, L.; Ngo, P. Political uncertainty and accounting conservatism. *Eur. Account. Rev.* **2021**, *30*, 277–307. https://doi.org/10.1080/09638180.2020.1760117.
- 58. Chen, Q.; Goldstein, I.; Jiang, W. Price informativeness and investment sensitivity to stock price. *Rev. Financ. Stud.* **2006**, *20*, 115–147. https://doi.org/10.1093/rfs/hhl024.
- 59. Edmans, A.; Jayaraman, S.; Schneemeier, J. The source of information in prices and investment-price sensitivity. *J. Financ. Econ* **2017**, *126*, 76–96. https://doi.org/10.1016/j.jfineco.2017.06.017.
- 60. Foucault, T.; Frésad, L. Cross-listing, investment sensitivity to stock prices, and the learning hypothesis. *Rev. Financ. Stud.* **2012**, 25, 3305–3350. https://doi.org/10.1093/rfs/hhs093.
- 61. Marhfor, A.; Bouslah, K.; M'Zali, B. Corporate social responsibility as a managerial learnings process. *ACRN J. Financ. Risk Perspect.* **2021**, *10*, 77–94. https://doi.org/10.35944/jofrp.2021.10.1.005.
- 62. Luo, W.; Wang, Y.; Zhang, X. Monetary policy uncertainty and firm risk-taking. *Pac. Basin Financ. J.* 2022, 71, 101695. https://doi.org/10.1016/j.pacfin.2021.101695.
- 63. Zhang, C.; Yang, C.; Liu, C. Economic policy uncertainty and corporate risk-taking: Loss aversion or opportunity expectations. *Pac. Basin Financ. J.* **2021**, 39, 101640. https://doi.org/10.1016/j.pacfin.2021.101640.
- 64. McWilliams, A.; Siegel, D. Corporate social responsibility: A theory of the firm perspective. *Acad. Manag. Rev.* **2001**, *26*, 117–127. https://doi.org/10.2307/259398.
- Maon, F.; Lindgreen, A.; Swaen, V. Organizational Stages and Cultural Phases: A Critical Review and a Consolidative Model of Corporate Social Responsibility. *Intl. J. Manag. Rev.* 2010, 12, 20–38. https://doi.org/10.1111/j.1468-2370.2009.00278.x.
- 66. Farcane, N.; Deliu, D.; Bureană, E. A Corporate Case Study: The Application of Rokeach's Value System to Corporate Social Responsibility (CSR). *Sustainability* **2019**, *11*, 6612. https://doi.org/10.3390/su11236612.
- 67. Dutton, J.E.; Roberts, L.M.; Bednar, J. Pathways for Positive Identity Construction at Work: Four Types of Positive Identity and the Building of Social Resources. *Acad. Manag. Rev.* **2010**, *35*, 265–293. https://doi.org/10.5465/amr.35.2.zok265.
- 68. Porter, M.E.; Kramer, M.R. The link between competitive advantage and corporate social responsibility. *Harv. Bus. Rev.* **2006**, 84, 78–92. https://doi.org/10.1108/sd.2007.05623ead.006.
- 69. Bouslah, K.; Hmaittane, A.; Kryzanowski, L.; M'Zali, B. CSR Structures: Evidence, Drivers, and Firm Value Implications. *J. Bus. Ethics* 2022. https://doi.org/10.1007/s10551-022-05219-6.
- 70. Jo, H.; Harjoto, M.A. The causal effect of corporate governance on corporate social responsibility. *J. Bus. Ethics* **2012**, *106*, 53–72. https://doi.org/10.1007/s10551-011-1052-1.
- 71. Vishwanathan, P.; Van Oosterhout, H.; Heugens, P.P.M.A.R.; Duran, P.; Van Essen, M. Strategic CSR: A concept building meta-analysis. *J. Manag. Stud.* **2020**, *57*, 314–350. https://doi.org/10.1111/joms.12514.
- 72. Cui, J.; Jo, H.; Na, H. Does corporate social responsibility affect information asymmetry? *J. Bus. Ethics* **2018**, *148*, 549–572. https://doi.org/10.1007/s10551-015-3003-8.
- 73. Cormier, D.; Magan, M. The impact of social responsibility disclosure and governance on financial analysts' information environment. *Corp. Gov.* **2014**, *14*, 467–484. https://doi.org/10.1108/CG-01-2013-0012.
- 74. Montiel, I.; Husted, B.; Christmann, P. Using private management standard certification to reduce information asymmetries in corrupt environments. *Strateg. Manag. J.* **2012**, *33*, 1103–1113. https://doi.org/10.1002/smj.1957.
- 75. Ramchander, S.; Schwebach, R.G.; Staking, K. The informational relevance of corporate social responsibility: Evidence from DS400 index reconstitutions. *Strateg. Manag. J.* **2012**, *33*, 303–314. https://doi.org/10.1002/smj.952.
- Refinitiv. Environmental, Social and Governance (ESG) Scores from Refinitiv. 2022. Available online: https://www.refinitiv.com/content/dam/marketing/en_us/documents/methodology/refinitiv-esg-scores-methodology.pdf (accessed on 15 June 2022).